Author's response to reviews

Title: Accuracy of self-reported body weight, height and waist circumference in a Dutch overweight working population

Authors:

Johanna C Dekkers (c.dekkers@vumc.nl)
Marieke F van Wier (m.vanwier@vumc.nl)
Ingrid JM Hendriksen (ingrid.hendriksen@tno.nl)
Jos WR Twisk (j.w.r.twisk@vumc.nl)
Willem van Mechelen (w.vanmechelen@vumc.nl)

Version: 4 Date: 15 September 2008

Author's response to reviews:

September 15, 2008

Re: Resubmission of the manuscript ‘Accuracy of self-reported body weight, height and waist circumference in a Dutch overweight working population’

Dear editor,

Please find enclosed a revised version of the manuscript “Accuracy of self-reported body weight, height and waist circumference in a Dutch overweight working population” by J Caroline Dekkers et al.

We are delighted that, in principle, our manuscript has been accepted for publication in your. We have improved the manuscript by addressing the comments made by reviewer 1 as well as the comments made by the editor.

According to the comments made by the reviewer, we added references 37 and 38 to the manuscript, which were inadvertently missing. Also, we changed ‘sex’ into ‘Sex’ in Table 6a and 6b.

According to the comments made by the editor, we shortened the abstract of the manuscript to a maximum of 350 words and removed the Table placement text.

We are looking forward to publication of our manuscript in your journal.

Yours sincerely,

Caroline Dekkers, PhD
EMGO Institute and Department of Public and Occupational Health
VU University medical center,
Van der Boechorststraat 7,
1081 BT Amsterdam, The Netherlands
Phone: +31 (20) 444 8206
Fax: +31 (20) 444 8387
e-mail: c.dekkers@vumc.nl